Author’s response to reviews

Title: Validation of a Susceptibility, Benefits, and Barrier Scale for Mammography Screening among Peruvian Women: a Cross-Sectional Study

Authors:

Moises A Huaman (mhuamanjoo@gmail.com)
Kelly I Kamimura-Nishimura (kellykamimuranishimura@gmail.com)
Mariano Kanamori (marianokanamori@hotmail.com)
Alejandro Siu (alejandrosiu@upch.edu.pe)
Andres G Lescano (Willy.lescano@med.navy.mil)

Version: 4 Date: 9 November 2011

Author's response to reviews: see over
November 08, 2011

BMC Women’s Health Editors

Dear Editors,

We are submitting a revised version of the manuscript titled “Validation of a Susceptibility, Benefits, and Barrier Scale for Mammography Screening among Peruvian Women: a Cross-Sectional Study”.

For the manuscript preparation we followed the instructions provided by your web page and we accept the conditions posed. This manuscript is original and has not been published elsewhere nor submitted for publication simultaneously.

We have incorporated all of formatting changes requested into a revised manuscript (major revisions). Do not hesitate to contact us if you require further information.

Sincerely,

Moises A. Huaman, MD
Universidad Peruana Cayetano Heredia, Lima, Peru
E-mail: mhuamanjoo@gmail.com